Author’s response to reviews

Title: Does medical training in Thailand prepare doctors for work in community hospitals? An analysis of critical incidents

Authors:

Dumrongrat Lertrattananon (dumrongrat.ler@mahidol.edu)
Wirun Limsawart (lwirun@gmail.com)
Alan Dellow (alan.dellow@btinternet.com)
Helen Pugsley (pugsleyh3@cardiff.ac.uk)

Version: 2 Date: 08 Jul 2019

Author’s response to reviews:

Reviewer 1

1. I have substituted "shortage" for "dilemma". Dilemma is a rather emotional word and raises questions about exactly what the dilemma is. Shortage is factually correct and precise.

2. I have added an explanation for normal and rural tracks. However, this research did not aim to compare these tracks.

Reviewer 2

The manuscript has been grammar checked by a native English speaker (Dr. Alan Dellow) who has co-authored several medical research papers and books.